Author’s response to reviews

Title: Adjuvant chemotherapy could not bring survival benefit to HR-positive, HER2-negative, pT1b-c/N0-1/M0 invasive lobular carcinoma of the breast: a propensity score matching study based on SEER database

Authors:

guangfu hu (guangfu12@fudan.edu.cn)

Guangxia Hu (940502551@qq.com)

Chengjiao Zhang (1010647455@qq.com)

Xiaoyan Lin (linxiaoyan077@163.com)

Ming Shan (supergoo99@163.co)

Yanmin Yu (yuyanmin75@126.com)

Yongwei Lu (763555862@qq.com)

Ruijie Niu (twinkleniu@163.com)

Hui Ye (yehui1969@126.com)

Cheng Xu (xucheng@live.cn)

Cheng Wang (wangcheng1212@hotmail.com)

Version: 1 Date: 21 Sep 2019

Author’s response to reviews:

Object: BCAN-D-19-02948

Adjuvant chemotherapy could not bring survival benefit to HR-positive, HER2-negative, pT1b-c/N0-1/M0 invasive lobular carcinoma of the breast: a propensity score matching study based on SEER database

Thank you for consideration of our manuscript for publication in your journal.
We have added the description of the role of funding in the manuscript. We have modified the supplementary files and uploaded the new files in version 2 of our manuscript according to editors’ comments.